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Highlights:

A major activity for the group this week was a design review of the Florida State University Big Light FEL.  We also supported the JLab Facility Safety Analysis Document and Safety Envelope review through interviews with several members of the FEL staff.   
Management:
We prepared for an all day review of the FSU FEL design.   FSU and Daresbury scientists spent Thursday and Friday with us analyzing and commenting on the design.  The design will be converted during August to a NSF proposal submission.
 

New DOE safety orders require modifications to our facility safety documentation, operational safety envelopes, and accelerator readiness process.  We hosted an external group this week to determine what changes are required for us to meet the new DOE orders.   

We prepared several proposals to the JSA fund to support tests on the FEL.  We also discussed possible proposals to the JTO for the current call and discussed efforts that might be submitted to DOE.

We hosted a visit from Delegate A. Howell of the Science and Education Committee and in a separate visit hosted the Virginia Economic Development Partnership

Gwyn Williams attended a THz optics and facilities review at Daresbury Lab, and presented a summary of the JLab programs.
Tech Transfer:   

We held the monthly TRC meeting.

We participated in the monthly Hampton Roads Research Partnership (HRRP) Board of Directors meeting.
We completed the draft of the MOU between HRRP and JLab. Document is in signature phase.
Injector:

The remote stalk retractor mechanism for remote re-cesiation is being tested on the FEL gun after the bellows flange was machined for mounting the motor and limit switch brackets. 

GTS 

Don Bullard has completed the polishing of the Rowoski profiles on the anode plate. It is now ready for final polishing and coating. Parts for the GTS clean-room are in order and we will start assembly early next week. 

The GTS gun chamber vacuum firing was a complete success. This was the first time the FEL group attempted such a process. The one-week ramp up to 400 C and the one week soak at that temperature went flawlessly. We made a convection oven with 6kW electrical heaters instead of the usual hot air blowers. This choice proved to be an excellent one and we will implement it for future FEL gun bakeouts. After the chamber had cooled down to room temperature, the ion pump is reading about 20 nA of current, so the pressure is in the low 1E-10 Torr. Phil Adderley and Marcy Stutzman from the Source group will perform out-gassing measurements of our 316 LN SS chamber for comparison with their 304 L SS chamber which was vacuum fired with a similar process. 
Carlos Hernandez-Garcia presented the invited talk entitled "Vacuum and surface physics of high average current photoemission electron guns" at the 17th International Vacuum Conference held in Stockholm Sweden. The talk was extremely well received.
Instrumentation and Controls:

Effort continued on the GTS HVPS Analog card. The front panel has been designed and the backplane schematic was modified. The PCB layout has begun.  Modifications to the HV pulse generator for the GTS Drive Laser were completed. Installation continued on the SF6 system for the Gun Test Stand.  The 0.25" aluminum sheet has been hung in the vault for system mounting. Layout was done on the sheet for installation guidance. The SF6 compressor has been hung flush on the plate with anchors into the wall. The GTS drive laser table mount holes have been modified to accept sleeves.

   Work continues on the LPSS UL upgrades. Debugging in UL1 revealed a bad PLC output module and has been replaced.  New sweep boxes were installed in UL 5.  The LPSS documentation pdf's have been posted onto the wiki and have made some additional updates. 

(https://fel.jlab.org/wiki/index.php/LPSS_Documentation).

Some general FEL system maintenance activities took place this week.  Power supply fan filters were replaced in the gallery. Some rack chassis's had the accumulated dust vacuumed out. Additional cabinets and equipment were moved out of UL 5 to make way for new installations.

Beam Transport:

Improvements and Upgrades
Gun Test Stand (GTS)

•
The design package per ASME Code Section VIII design standards for the SF6 Tank system remains in peer review by Robby Hicks.  He is also performing the finite element calculation on the intersection with the gun vacuum tank using ANSYS Workbench.  The drawings are complete and are awaiting Robby’s signature.  Preliminary copies with partial signature have been given to the shop. The fabrication specification that will satisfy the requirements of the JLab pressure vessel committee for maintaining records of material certification, the welding per a qualified procedure and the maintenance of a material tracking system remain in preparation. The half shells are being prepared for seam welding.  The weld samples to qualify the procedure are at Advex Co. for testing.  The flange ring material was received on Monday. 

•
The Gun Solenoid remains in fabrication at New England Technicoil.

•
Matthew Marchlik and Brian Hankla continued work on the SF6 Gas Transfer System.  They did a final review.  Installation began.

•
The simplified Brewster window housing is in fabrication.  

•
The parts for hanging the Drive Laser Table system were put into the JLab Shop for fabrication.

SF Sextupole Magnet Testing 
•
George Neil determined that enough money was available in our budget to start testing one Sextupole with the new system.  Jim Coleman is arranging the details of starting this test up. 

Magnetic anomalies 
•
No work this week
National High Field Magnet Laboratory FEL

•      I was put together the magnet talk for the review taking place today.

Optical Transport/User Labs
Installation of the User transport equipment commenced this week. The high power mirror cassette chamber was aligned and installed in Lab 5. We should be able to do likewise for Lab 3a by the end of the day. The two high power mirror cassettes have now been plumbed for cooling. We have received all of the new spool section to complete the lab transport. The bellows sections that attach to the cassette chambers have not yet arrived to complete the optical transport. Detailing the machine drawings for the high power turning mirror assemble (pictured below) is ongoing.
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Terahertz: 
Discussions took place with Peter Gardner, Mark Surman, Peter Weightman and Hannan Fersi regarding their continued user experiments in the bio-medical area.  We also discussed some planned spectroscopy experiments at Brookhaven with Peter G, Hannan and possibly Mark next month, because of the extended reach compared with the THz Test Facility at Daresbury.  These experiments will primarily be on proteins, and the relevance to our own programs would be in wound healing.  Meanwhile we continued with preparations for our magneto-optical experiments, also next month at Brookhaven, in collaboration with Los Alamos and Florida State.
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